Author’s response to reviews

Title: The Diagnostic and Prognostic Utility of Risk Factors Defined by the AHA/ACCF on the Evaluation of Cardiac Disease in Liver Transplantation Candidates

Authors:

Sarah Alexander (sarahalexander1@gmail.com)
Meron Teshome (merkedeme@yahoo.com)
Hena Patel (hena_n_patel@rush.edu)
Edie Chan (edie_y-chan@rush.edu)
Rami Doukky (rami_doukky@rush.edu)

Version: 1 Date: 08 Mar 2019

Author’s response to reviews:

Dear Dr. Mockridge:

Thank you very much for the opportunity to revise and resubmit our manuscript entitled: “The Diagnostic and Prognostic Utility of Risk Factors Defined by the AHA/ACCF on the Evaluation of Cardiac Disease in Liver Transplantation Candidates” for publication in BMC Cardiovascular Disorders. In the revised manuscript we address point-by-point every comment or question raised by the reviewer. To facilitate the review process, we provided two versions of the manuscript, a version labeled “CHANGES” that highlights our changes in red text and a “CLEAN” version for final production. We hope that you find our revised manuscript suitable for publication in BMC Cardiovascular Disorders.

We are looking forward to hearing back from you.

Sincerely,

Rami Doukky, MD, MSc, FACC